Author's response to reviews

Title: Family practice nurses supporting self-management in older patients with mild osteoarthritis: a randomized trial

Authors:

Raymond Wetzels (r.wetzels@kwazo.umcn.nl)
Chris van Weel (c.vanweel@hag.umcn.nl)
Richard Grol (r.grol@kwazo.umcn.nl)
Michel Wensing (m.wensing@kwazo.umcn.nl)

Version: 4 Date: 19 January 2008

Author's response to reviews:

Dear Sirs,

We have adjusted the formatting changes you requested:

- We removed the job titles from the authors list and placed the authors list on one line
- We included a Conclusions section after the Discussion
- We included a 'Competing interests' section between the Conclusions and Authors' contributions
- We moved the Authors contributions to above the acknowledgments
- We changed Contributions authors - to Authors' contributions
- We included the funding information in the acknowledgments
- We have included in the methods section the justification of why our trial is not in a publicly accessible registry (see attach).

Hopefully we addressed the necessary formatting changes to your satisfaction.

Yours sincerely, Raymond Wetzels